Calculus I Fall 2001 MA113:13-15, 19-21 Russell Brown Below are the remaining homework assignments before the third test. The third exam will be on Tuesday, 13 November 2001.

• Homework #11. §3.5 #10, 14, 54, 66a) §3.6 #8 (Long!) §3.8 #10, 14. Due Wednesday, 7 November 2001.

• Homework F. §3.8 # 20. Due Friday, 9 November 2001.

• Homework #12. §3.10 #28, 56. §4.1 # 4, 20, 22. Homework #12 will not be collected for grading–I doubt the grader would be able to return it before the exam. I will hand out solutions to Homework #12 on Friday, 9 November.

October 26, 2001